Author's response to reviews

Title: The Health Informatics Trial Enhancement Project (HITE): Using routinely collected primary care data to identify potential participants for a depression trial

Authors:

   Joanna Mcgregor (J.I.McGregor@swansea.ac.uk)
   Caroline Brooks (c.brooks@swansea.ac.uk)
   Padmaja Chalasani (Padmaja.Chalasani@cwp.nhs.uk)
   Jude Chukwuma (j.chukwuma@swansea.ac.uk)
   Hayley Hutchings (H.A.Hutchings@swansea.ac.uk)
   Ronan A Lyons (r.a.lyons@swansea.ac.uk)
   Keith Lloyd (k.r.lloyd@swansea.ac.uk)

Version: 7 Date: 12 April 2010

Author's response to reviews:

Further to your email of 31st march 2010, we have made the following revisions -
1. reference 15 has been cited at page 6 line 2
2. reference 19 has been deleted
3. Subsequent references (19-32) have been reordered and renumbered in the text and reference list.
4. Spelling has been checked

Many thanks